Author's response to reviews

Title: A Protocol for a Systematic Review of Knowledge Translation Strategies in the Allied Health Professions

Authors:

Shannon D Scott (shannon.scott@ualberta.ca)
Lauren Albrecht (lauren.albrecht@nurs.ualberta.ca)
Kathy O'Leary (kathy.o'leary@ualberta.ca)
Geoff DC Ball (geoffball@med.ualberta.ca)
Donna M Dryden (ddryden@ualberta.ca)
Lisa Hartling (lisa.hartling@ualberta.ca)
Anne Hofmeyer (anne.hofmeyer@unisa.edu.au)
C Allyson Jones (Allyson.jones@ualberta.ca)
Kathy Kovacs Burns (kathy.kovacsburns@ualberta.ca)
Amanda S Newton (mandi.newton@ualberta.ca)
David S Thompson (dthompson@nosm.ca)
Terry P Klassen (tklassen@mich.ca)

Version: 4 Date: 27 May 2011

Author's response to reviews: see over
May 27, 2011

Martin Eccles & Brian Mittman
Editors-in-Chief
Implementation Science
c/o BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Dear Mr. Eccles & Mr. Mittman,

It is with pleasure that we submit a revised manuscript entitled “A Protocol for a Systematic Review of Knowledge Translation Strategies in the Allied Health Professions” for publication consideration in Implementation Science. All additional revisions have been completed, including title, author list, formatting, in-text citations, reference list website, additional files, etc. in order to meet Implementation Science guidelines.

Please do not hesitate to contact me if you have any questions regarding our submission. We look forward to your review of our work. Correspondence regarding this article should be addressed to Shannon D. Scott, 3rd floor Clinical Sciences Building, Faculty of Nursing, University of Alberta, Edmonton, Alberta, Canada T6G 2G3.

Most sincerely,

Shannon D. Scott, RN, PhD
Assistant Professor, Faculty of Nursing, University of Alberta